From wendy.adams@comlab.ox.ac.uk Tue Feb  6 12:54:16 2007
Date: Tue, 06 Feb 2007 12:54:25 +0000
From: Wendy Adams <wendy.adams@comlab.ox.ac.uk>
To: rhb@maths.ox.ac.uk, wilkie@maths.ox.ac.uk, zilber@maths.ox.ac.uk, hap@maths.ox.ac.uk, Samson.Abramsky@comlab.ox.ac.uk, henke@maths.ox.ac.uk, james.martin@stats.ox.ac.uk, Luke.Ong@comlab.ox.ac.uk, Victor Flynn <flynn@maths.ox.ac.uk>, Bob.Coecke@comlab.ox.ac.uk, georg.gottlob@comlab.ox.ac.uk
Subject: Exam Header

Dear all,

Enclosed is the exam header you should use to prepare your mini-project 
this term for the MSc in Mathematics and Foundations of Computer Science 
Course.

Wendy

\documentclass[12pt,a4]{article}

\begin{document}

\begin{center}
{\bf The University of Oxford}\\[5mm]

{\bf MSc (Mathematics and Foundations of Computer Science)}\\[5mm]

{\large\bf Your Subject}\\[3mm]

{\bf Hilary Term 2007}

\end{center}

\medskip

\noindent {\em Answer as many problems as you can.}

\medskip
\noindent {\em Please write on one side of the paper only.}\\[3ex]
\smallskip
\noindent
\end{document}

-- 
Wendy Adams
Academic Administration
Oxford University Computing Laboratory
Wolfson Building
Parks Road
Oxford OX1 3QD
Tel: 01865 283559
Fax: 01865 273839
Email: wendy.adams@comlab.ox.ac.uk
